Author’s response to reviews

Title: Indicator-Based Public Health Monitoring in Old Age in OECD Member Countries: A Scoping Review

Authors:

Maike Miriam Grube (maike.grube@posteo.de)
Ralph Möhler (moehler@ifem.uni-freiburg.de)
Judith Fuchs (fuchsj@rki.de)
Beate Gaertner (gaertnerb@rki.de)
Christa Scheidt-Nave (scheidt-navec@rki.de)

Version: 4 Date: 04 Jul 2019

Author’s response to reviews:

1. Figure 2 is included but is not cited or presented in the results section. Please refer to Figure 2 in the text of the results section.

Answer: Many thanks for pointing this out. We included a reference to Figure 2 in the results section (page 9, line 188).

2. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Answer: We apologize and deleted the remaining track changes from the manuscript.